Author’s response to reviews

Title: Evaluating the feasibility of a nurse-led self-management support intervention for kidney transplant recipients: a pilot study

Authors:

Janet Been-Dahmen (j.m.j.been-dahmen@hr.nl)
Denise Beck (d.beck@erasmusmc.nl)
Mariëlle Peeters (m.a.c.peeters@hr.nl)
Heleen van der Stege (h.a.van.der.stege@hr.nl)
Mirjam Tielen (m.tielen@erasmusmc.nl)
Marleen van Buren (m.c.vanburen@erasmusmc.nl)
Erwin Ista (w.ista@erasmusmc.nl)
AnneLoes van Staa (a.van.staa@hr.nl)
Emma Massey (e.massey@erasmusmc.nl)

Version: 4 Date: 21 Feb 2019

Author’s response to reviews:

1. Copyright and use of logo’s: Figure 1 is not under copyright. Similarly as employees of these institutions we can use the respective institutional logos for such products. We chose not to copyright the Self-management Web so as to minimize barriers for dissemination, distribution and use. We have discussed removal of the institutional logo’s but have chosen include them so as to identify the institutions involved in development of the tool and to promote correct referencing of this paper.

2. Clean version: only the clean version was uploaded in the previous submission. We have checked the colour and track changes. As far as we can see there were no changes necessary. We have uploaded this checked version just in case.